Author's response to reviews

Title: A clinical trial to evaluate the effects of Plantago ovata husk in Parkinson patients: changes in levodopa pharmacokinetics and biochemical parameters

Authors:

Nelida Fernandez (nelida.fernandez@unileon.es)
Luis Hernandez (lhernandez@saludcastillayleon.es)
Matilde Sierra (matilde.sierra@unileon.es)
M Jose Diez (mjdiel@unileon.es)
Angela Calle (angela.calle@unileon.es)
Demetrio Carriedo (dcarriedo@saludcastillayleon.es)
Ana Sahagun (amsahp@unileon.es)
Anna Anguera (dftnfm@unileon.es)
Juan Jose Garcia (jjgarv@unileon.es)

Version: 2 Date: 3 July 2013

Author's response to reviews: see over
1. Trial registration where clinical trials are described: We notice that you are reporting a clinical trial but have not cited a trial registration number. The number has been included at the end of the abstract section.

2. Authors' information: Please place the Authors' Contributions section after Competing interests. Please check the instructions for authors on the journal website for the correct format to use for Authors' Contributions. The section has been included.

3. Competing interests: Manuscripts should include a Competing interests section. This should be placed after the Conclusions/Abbreviations. Please consider the following questions and include a declaration of competing interests in your manuscript: The section has been included.